Author's response to reviews

Title: Influences on Eating and Physical Activity Behaviors: A Qualitative Study of Adolescents in a Periurban Area in Lima, Peru

Authors:

Jinan C Banna (jcbanna@hawaii.edu)
Opal V Buchthal (opalb@hawaii.edu)
Treena Delormier (treenad@hawaii.edu)
Hilary M Creed-Kanashiro (hmcreed@iin.sld.pe)
Mary E Penny (mpenny@iin.sld.pe)

Version: 2
Date: 22 June 2015

Author's response to reviews:

The abstract has been added to the second page of the manuscript, and line and page numbers have been added throughout.